Author's response to reviews

Title: The Prevalence of Exposure to Domestic Violence and the Factors Associated with Co-occurrence of Psychological and Physical Violence Exposure: a Sample from Primary Care Patients

Authors:

Polona Selic (polona.selic@siol.net)
Katja Pesjak (katja.pesjak@mf.uni-lj.si)
Janko Kersnik (janko.kersnik@ozg-kranj.si)

Version: 6 Date: 3 August 2011

Author's response to reviews: see over
Cover letter

Title:

The Prevalence of Exposure to Domestic Violence and the Factors Associated with Co-occurrence of Psychological and Physical Violence Exposure: a Sample from Primary Care Patients

Formatting changes made:

Major revisions

1. Tables: Authors were asked to remove the tables from the main text of the manuscript and place them in a tables section after the references.

All tables were removed from the main text of the manuscript and placed after the references, in a tables section. We checked that that all tables were cited in a numerical order.

Minor revisions

2. Abbreviations: Authors were asked to format abbreviations in to a paragraph and to make sure all abbreviations are spelt out and occur in alphabetical order.

Abbreviations were formatted in to a paragraph and listed in alphabetical order. The spelling was checked. We hope that we understood correctly what we were supposed to do with abbreviations section.

3. Figures: We were warned that it was important for the final layout of the manuscript that the figures were cropped as closely as possible to minimise white space around the image.

We used the online figure guide and tried to follow instructions on: http://www.biomedcentral.com/info/ifora/figures.

We are somehow in doubt about the success of this revision. Therefore, we ask the Editorial team to let us know if the Figure is cropped as it was expected.